Author’s response to reviews

Title: The Discharge Communication Study: research protocol for a mixed methods study to investigate and triangulate discharge communication experiences of patients, GPs, and hospital professionals, alongside a corresponding discharge letter sample

Authors:

Katharine Weetman (K.weetman@warwick.ac.uk)
Jeremy Dale (jeremy.dale@warwick.ac.uk)
Emma Scott (e.j.scott@warwick.ac.uk)
Stephanie Schnurr (s.schnurr@warwick.ac.uk)

Version: 3 Date: 30 Sep 2019

Author’s response to reviews:

Thank you for your comments. The editor comment regarding author contributions has now been addressed and this section wording has been revised as follows:

"Authors’ contributions: KW drafted the initial manuscript and project design. JD, SS, and ES substantially contributed toward the conception and development of the protocol and the project design. All authors read and critically reviewed the manuscript. JD, SS and ES supervised the study data collection. All authors are accountable for all aspects of the work. All authors approved the final manuscript in its current state."